Author's response to reviews

Title: Estimation of cancer incidence and mortality attributable to alcohol drinking in China

Authors:

Hao Liang (lianghaopumc@gmail.com)
Jian-Bing Wang (wangjianbing1980@yahoo.com)
Hui-Juan Xiao (xhi60000@163.com)
Ding Wang (wendyding@126.com)
Wen-Qiang Wei (weiwg@cicams.ac.cn)
Paolo Boffetta (paolo.boffetta@i-pri.org)
You-Lin Qiao (qiaoy@cicams.ac.cn)

Version: 5 Date: 10 November 2010

Author's response to reviews: see over
Dear Dr. Roxane Rajabi,

Thank you very much for your letter dated September 23 and for the accompanying reviewers’ comments on our paper (9740505714254879). We are submitting a revised manuscript that incorporates the comments of the reviewers. A point-by-point response to 1 of the reviewers’ comments is attached as follows. Also, we have changed the format of our paper.

We hope that our paper has been revised satisfactorily and will be accepted for publication in BMC Public Health.

Sincerely,

Youlin Qiao    M.D. Ph.D.
Hao Liang       M.D.

Attachment:  Point-by-point response to the reviewer’s comments received on 11/10/2010

Reviewer #1:
Comment #1.1: In Methods (text) "for your sensitivity analysis/results you should describe if you have used specific software...."

Response 1.1: Actually, the sensitivity analysis we conducted did not use specific software, we use the same formula which include same RR, but different prevalence of alcohol drinking in 2002, to calculate PAF and cancer burden.